Author’s response to reviews

Title: Discovery biology of neuropsychiatric syndromes (DBNS): a center for integrating clinical medicine and basic science

Authors:

ADBS consortium (adbs.project@gmail.com)
sanjeev jain (sjain.nimhans@gmail.com)

Version: 2 Date: 04 Mar 2018

Author’s response to reviews:

From,  
Prof Sanjeev Jain  
BPSY-D-17-00554  
NIMHANS

To  
Prof. Benjamin Ragen  
The editor  
BMC Psychiatry

Dear Sir,

Sub: Our manuscript BPSY-D-17-00554 titled “Discovery biology of neuropsychiatric syndromes (DBNS): a centre for integrating clinical medicine and basic science”

With reference to editors comments on our article, the following documentations/ changes are provided.

Editor's comments:

1) Please include the email addresses for all authors.

Response: This has been done. Please note that the consortium has two additional members who have been added in the list (Sydney Moirangthem, Vijay Kumar K.G.). The additional authors have also read and approved the current manuscript. The NIMHANS ethics committee has also reviewed and approved this change in the proposal. Kindly allow this addition in the manuscript.

2) In the ethics section please specify all three institutions.
Response: We have added this information in ethics section as per suggestion.

3) Please use numbers to indicate authors' affiliations instead of "@, #, *
Response: The author affiliations have been numbered as suggested.

Sincerely,

Sanjeev Jain